

Young Adult Male Patients With Childhood-onset IBD Have Increased Risks of Compromised Cortical and Trabecular Bone Microstructures

Gudmundur Vignir Sigurdsson, MD,^{*,†,} Susanne Schmidt, MD, PhD,[‡] Dan Mellström, MD,^{§,} Claes Ohlsson, MD,^{§,¶} Robert Saalman, MD,^{*} and Mattias Lorentzon, MD^{§,1**,}

^{*}Department of Pediatrics, Institute of Clinical Sciences, The Sahlgrenska Academy at the University of Gothenburg, and Queen Silvia's Children's Hospital, Gothenburg, Sweden

[†]Heilbrigdisstofnun Sudurlands, Selfoss, Iceland

*Labcorp Clinical Development SARL, Rueil-Malmaison, France

[§]Sahlgrenska Osteoporosis Centre, Centre for Bone and Arthritis Research, Department of Internal Medicine and Clinical Nutrition, Institute of Medicine, The Sahlgrenska Academy at the University of Gothenburg, Gothenburg, Sweden

¹Region Västra Götaland, Sahlgrenska University Hospital, Department of Drug Treatment, Gothenburg, Sweden

Region Västra Götaland, Geriatric Medicine, Sahlgrenska University Hospital, Gothenburg, Sweden

**Mary MacKillop Institute for Health Research, Australian Catholic University, Melbourne, Australia

Address correspondence to: Gudmundur Vignir Sigurdsson, Heidarbrun 32, 810 Hveragerdi, Iceland (vignir.sigurdsson@gu.se).

Background: Young adults with childhood-onset inflammatory bowel disease (IBD) have increased risks of low areal bone mineral density and low skeletal muscle mass. Volumetric BMD (vBMD), bone geometry and microstructures, in addition to possible associations with skeletal muscle index (SMI) and physical exercise have been scarcely studied in this patient group.

Patients and methods: In total, 49 young adult male patients with childhood-onset IBD and 245 age- and height-matched young adult male controls were scanned with high-resolution peripheral quantitative computed tomography. Bone geometry, vBMD, and bone microstructures were calculated as median values and compared between the patients and controls. Multivariable linear regression analyses were performed to determine the independent associations among IBD diagnosis, SMI (kg/m²), and physical exercise.

Results: The group of young adult patients had, in comparison with the controls, significantly smaller median cortical area (126.1 mm² vs151.1 mm², P < .001), lower median total vBMD (296.7 mg/cm³ vs 336.7 mg/cm³, P < .001), and lower median cortical vBMD (854.4 mg/ cm³ vs 878.5 mg/cm³, P < .001). Furthermore, the patients compared with the controls had lower median trabecular volume fraction (16.8% vs 18.2%, P < .001) and thinner median trabeculae (0.084 mm vs 0.089 mm, P < .001). The differences between the patients with IBD and controls persisted in multivariable analyses that included adjustments for SMI and physical exercise.

Conclusions: Young adult men with childhood-onset IBD are at increased risk of having reduced bone quality in both the cortical and trabecular bone structures compared with normative matched controls.

Lay Summary

Young adult men with childhood-onset IBD appear to have deficits in both cortical and trabecular bone microstructures, measured with high resolution peripheral computed tomography, compared with age- and height-matched young adult male controls. **Keywords:** bone mineral density, HR-pQCT, IBD, SMI, physical exercise

Introduction

Patients with inflammatory bowel disease (IBD) are at high risk of low areal bone mineral density (aBMD).¹⁻¹³ This risk has been reported for children,¹⁻⁴ young adults with childhood-onset IBD,⁶⁻⁸ and adults.⁹⁻¹³ Most of the data regarding the bone health of patients with IBD are derived from dual x-ray absorptiometry (DXA) measurements. Dual x-ray absorptiometry is considered the gold standard for assessing aBMD in clinical practice. There is a strong association between low aBMD and increased risk of fracture in the adult population.^{14,15} However, the DXA method suffers from the limitation that it does not provide information about the 3D bone structure (volumetric BMD [vBMD]), bone geometry, or bone microstructure characteristics. These bone structures encompass the dense outer cortical layer and the honeycomb-like trabecular structure of the medullary cavity.

In recent years, high-resolution peripheral quantitative computed tomography (HR-pQCT) has become available with a spatial resolution of 82 µm. In addition to allowing more accurate measurements of bone geometry and vBMD, HR-pQCT can measure the cortical and trabecular bone microstructures that make up the structural architecture of bone. These estimations of the structural architecture provide additional important clinical information. For instance,

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/ licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For

commercial re-use, please contact journals.permissions@oup.com

Received for publication: March 11, 2022. Editorial Decision: August 1, 2022

^{© 2022} Crohn's & Colitis Foundation. Published by Oxford University Press on behalf of Crohn's & Colitis Foundation.

Key Messages

What is already known?

Young adults with childhood-onset IBD have an increased risk of low areal bone mineral density.

What is new here?

We show that in young men, childhood-onset IBD is associated with a deficit in both cortical and trabecular bone.

How can this study help patient care?

With this in mind, preventive and supportive measures could be taken in childhood and young adulthood to promote better bone health in the patient's future.

a few studies have revealed that HR-pQCT-derived bone parameters, such as cortical area and mass, predict fractures in older men—independently of aBMD.^{16,17}

To our knowledge, only 2 HR-pQCT studies have been conducted with patients who have IBD, and those studies have reported somewhat conflicting results. Pepe and colleagues examined adolescents and young adults (mean age 23 years; range, 12-33 years) with IBD and found deficits in their trabecular microstructures without significant changes in their cortical structures.¹⁸ However in a study of adults with IBD (median age 44 years; interquartile range [IQR], 31-54 years), Haschka and colleagues described deficits primarily in the cortical microstructures. In addition, they reported deficits in trabecular structures only in patients with Crohn's disease.¹⁹

A large proportion of patients with IBD have a low skeletal muscle index (SMI; ie, the weight of lean mass in the arms and legs divided by the height squared, kg/m²)²⁰ and engage in physical exercise only to a limited extent.²¹ Deficits in SMI²⁰ and lack of physical exercise²² are associated with compromised BMD in young adult patients with childhoodonset IBD. Previously published studies involving HR-pQCT analyses have not studied how SMI and physical exercise are associated with the bone microstructure deficits seen in patients with IBD.

This study aimed to investigate the extent of microstructural alterations in young adult males with IBD and the association between these changes and the patient's SMI and the amount of physical exercise. A cohort of 49 young adult male patients, all with childhood-onset of IBD, was scanned with HR-pQCT and compared with 249 normative controls, matched for age, gender, and height (ratio of 1:5).

Subjects and Methods

Study Participants

The present study is part of our prospective longitudinal project concerned with BMD and body composition in patients with childhood-onset IBD. Initially, from 2003 to 2005, 144 patients (93 boys and 51 girls) were included, representing the entire spectrum of IBD, from the greater Gothenburg area, Sweden.² In the first follow-up (2005-2007), a total of 126 patients (81 males and 45 females) participated. The current study is a part of the second follow-up (2012-2015) conducted in early adulthood, in which 49 men (age range, 18-27 years) were included. In this study

the term "early adulthood" refers to the age period of 18 to 27 years.

In the current study, 49 male patients with childhood-onset IBD were included at the time of the second follow-up. For comparison, we had access to a large contemporary control group of males who were randomly selected from the same region and who participated in the GOOD study, a previously collected large and population-based cohort study.²³ Five male control subjects from the GOOD study were matched to each male patient with IBD based on propensity score for both age and height. In total, 829 control subjects were available for age- and height-specific matching. A patient to control subject ratio of 1:5 was chosen to ensure appropriate power and minimal variation of the matching variables. Females with childhood-onset IBD were not included in this study because we did not have access to a female control cohort.

Anthropometric, DXA, and body composition measurements for both the patients with childhood-onset IBD and the control subjects have previously been published as part of a larger cohort^{6,20} and are summarized in Table 1. Dual x-ray absorptiometry scans were used for estimation of areal BMD (g/cm²), bone mineral content (BMC, g) in the whole body, lumbar spine (L1–L4), total hip, and SMI. The DXA measurements were all performed at the Sahlgrenska University Hospital in Gothenburg (Sweden) with a Lunar Prodigy DXA (GE Medical Systems Lunar) in both patients and controls.

The disease-specific characteristics of the patient group has also been published previously^{6,20} and include IBD subcategory (ulcerative colitis, n = 32 [65.3%]; Crohn's disease, n = 17 [34.7%]); median (IQR) age at disease onset 12.3 (9.7-13.8) years; and median (IQR) disease duration of 11.1 (9.5-13.6) years. Disease extension was classified according to the Montreal classification.²⁴ The patients with ulcerative colitis had disease extension as follows: E1, n = 0 (0%); E2, n = 1 (3%); and E3, n = 31 (97%). The corresponding data on disease localization in patients with Crohn's disease were L1, n = 3 (18%); L2, n = 3 (18%); L3, n = 11 (65%); and L4, n = 5 (30%). The following classifications of disease behavior were noted: B1, n = 11 (65%); B2, n = 5 (29%); B3, n = 1(6%); and perianal disease, n = 2 (12%). Surgical intervention was reported in 14 patients (28.6%), including 9 patients with ulcerative colitis, all of whom underwent colectomy, and 5 patients with Crohn's disease, in whom small bowel surgery was performed.

All patients were treated over time according to current international guidelines from diagnosis to the time of bone microstructure measurement in young adulthood.^{25,26} A total of 32 (65.3%) patients had no disease symptoms at the time of the bone microstructure measurement. Current treatment regimens were as follows: prednisolone, n = 2 (4%); 5-aminosalicylic acid (5-ASA), n = 13 (26.5%); azathioprine, n = 18 (36.7%); antitumor necrosis factor (TNF)- α inhibitor, n = 2 (4.1%); methotrexate, n = 1 (2.0%); and no treatment, n = 12 (22.4%). The pharmacologic treatment of patients with childhood-onset IBD in our catchment area, as in the rest of Sweden, has followed the treatment principles stated in the National Care Program for IBD prepared on behalf of the Swedish Paediatric Society and the Swedish Society of Gastroenterology. The following pharmacologic treatments were prescribed from the time of diagnosis to the time of second follow-up: prednisolone, n = 47 (95.9%); 5-ASA, n = 49 (100%); azathioprine n = 38 (77.6%); and

	Patients With IBD $(n = 49)$	Age- and Height-Matched Controls $(n = 245)$	Difference (%)	Р
Age (years)	23.3 (21.6-25.2)	23.7 (23.5-24.3)	-1.7	.110
Height (cm)	178.3 (174.4-181.6)	178.3 (173.4-185)	0	.441
Weight (kg)	70.9 (63.3-79)	74.3 (67.9-82.3)	-4.6	.095
BMI (kg/m ²)	22.5 (20.8-24.4)	23.1 (21.2-25.1)	-2.5	.152
SMI (kg/m ²)	8.0 (7.3-8.7)	8.3 (7.7-9)	-3.6	.050
Physical exercise (hours/week)	1.0 (0-4)	1.4 (0-4)	-27.3	.408
Total body BMD (g/cm ²)	1.2 (1.1-1.3)	1.3 (1.2-1.3)	-6.3	<.001
Total body BMC (kg)	2.8 (2.6-3.4)	3.2 (2.9-3.5)	-10.6	<.001
Lumbar spine BMD (g/cm ²)	1.1 (1-1.2)	1.2 (1.1-1.3)	-7.8	<.001
L2L4 BMD (g/cm ²)	1.1 (1.0-1.2)	1.2 (1.1-1.3)	-11.8	<.001
L2L4 BMC (g)	51.3 (45.6-59.4)	58.2 (52.6-66.9)	-6.8	<.001
Femoral neck BMD (g/cm ²)	1.0 (0.9-1.1)	1.1 (1-1.2)	-5.7	.001
Femoral neck BMC (g)	5.4 (4.8-6.3)	6.0 (5.3-6.8)	-8.7	.002
Total hip BMD (g/cm ²)	1.0 (0.9-1.1)	1.1 (1-1.2)	-5.7	<.001

Table 1. Anthropometrics and DXA measurements in young adult males with childhood-onset inflammatory bowel disease compared with controls.

All values are presented as median (IQR). Difference between groupsweretested with Mann-Whitney U test.

Abbreviations: BMC, bone mineral content; BMD, bone mineral density; BMI, body mass index; IBD, inflammatory bowel disease; SMI, skeletal muscle index

TNF-α inhibitor as biologic therapy, n = 11 (22.4%). No biologic therapies other than TNF-α inhibitors were used in these patients. Physical exercise was defined as regular training and registered as hours per week, averaged for the entire previous year. Physical exercise data were used in regression analyses. Our research group recently published data regarding physical exercise habits in young adults with childhood-onset IBD.²²

We performed an analysis comparing the disease-specific characteristics of patients who participated in the second follow-up (participants, n = 49) with those of patients who did not choose to participate further in the study after the first follow-up (nonparticipants, n = 35). The following parameters did not differ significantly between the participants and nonparticipants: median age at diagnosis; median age; median disease duration; disease subcategory (ulcerative colitis vs Crohn's disease); surgical treatment; current corticosteroid use; current azathioprine use; and median BMD Z-scores.

High-resolution Peripheral Quantitative Computed Tomography

Measurements of bone geometry and microstructure (schematic illustration in Figure 1) with HR-pQCT were performed for all the study participants at the ultradistal tibia, ipsilateral to the nondominant arm, with the same Xtreme computed tomography (CT) scanner (Scanco Medical AG, Brüttisellen, Switzerland) using the manufacturer's standard in vivo protocol, as previously described.²⁷ A reference line was manually placed at the tibia's articular plateau with regular x-ray imaging guidance. The first CT slice was taken 22.5 mm distal to the reference line, and 104 parallel images were taken (82µm voxel size) in total, depicting a 9.02-mm 3D representation of the bone. Images were acquired over 2.8 minutes, and the effective dose was 3 uSv per measurement. Image quality was assessed, and the measurements were repeated until the quality was sufficient. A standard analysis was performed according to an earlier described protocol.²⁸

From the measurements, the following parameters were recorded: volumetric density (mg/cm³); cortical cross-sectional area (mm²); cortical volumetric BMD (mg/cm³); cortical thickness (mm); periosteal circumference (mm); trabecular cross-sectional area (mm²); trabecular bone volume fraction (%); trabecular number (mm⁻¹); trabecular thickness (mm); and trabecular separation (mm). The same 2 operators performed all the measurements and graded all the images for quality (score 1-5) using recommendations from the manufacturer.²⁹

Statistics

Statistical analyses were performed with the SPSS version 26 software (IBM Corp., Armonk, NY). Propensity score matching was performed with an R package plugin for SPSS. Continuous variables are presented as median (IQR). Categorical variables are presented as number (%). The differences between categorical and continuous variables between the 2 groups were tested with the Fisher exact test or the Mann-Whitney *U* test, respectively. Each multivariable linear regression analysis had one HR-pQCT measurement outcome as the dependent variable. Covariates were diagnosis of IBD, SMI (kg/m²), and physical exercise (hours/week). All tests were 2-tailed and conducted, assuming a significance level of 0.05.

Ethics

Informed written consent was obtained from all young adult patients with childhood-onset IBD and controls. The study was approved by the Regional Ethical Review Committee of the University of Gothenburg (Sweden; application numbers: for the IBD cohort 182-02 and 117-11; and the GOOD study cohort S600-02 and 017-08).

Results

In the present study, 49 young adult male patients with childhood-onset IBD and 245 age- and height-matched young

microstructures.

Figure 1. Illustration of bone showing basic trabecular and cortical

adult male controls were scanned with high-resolution peripheral quantitative computed tomography. A schematic illustration of the basic bone microstructures is illustrated in Figure 1.

Dual x-ray absorptiometry and anthropometric measurements have previously been published,^{6,20} and these results are described for the young adult male patients in Table 1. The median age of the patients was 23.3 years (IQR, 21.6-25.2) and 23.7 years (23.5-24.3) in controls. Out of the patients, 32 (65%) had ulcerative colitis, and 17 (35%) had Crohn's disease.

Bone Geometry and Microstructure Measurements in Young Adult Men With Childhood-onset IBD

The patient group with IBD had a median 16.5% smaller cortical area than the controls (P < .001). However, the patients and controls had similar total tibial and trabecular areas (Table 2). The median total vBMD was 12% lower in the patients with IBD than in the controls (P < .001), whereas the median cortical vBMD was 3% lower in the patients than in the controls (P < .001; Table 2).

The HR-pQCT measurements of bone microstructure revealed that the patients with IBD had 16.5% thinner median cortical layer than the controls (P < .001), but the periosteal circumference was similar in the patients and controls. Furthermore, the patient group had 7.7% lower median trabecular volume fraction than the controls (P < .001), whereas a 5.6% median reduction of the trabecular thickness was observed in the patients compared with the controls (P < .001). The patients showed a 7.0% greater median trabecular separation than the controls (P = .036). However, the number of trabeculae did not differ significantly between the groups (Table 2).

In order to evaluate differences in microstructure parameters between the 2 major IBD subcategories, 32 patients with ulcerative colitis and 17 with Crohn's disease were compared with 160 and 85 matched controls, respectively. The 2 IBD subcategories had similar deficits in bone geometry and bone microstructures compared with the matched controls—with 2 exceptions (Table 3). Median cortical BMD was significantly lower in the patients with ulcerative colitis (3.3% lower, P < .001) but not in the patients with Crohn's disease (1.7% lower, P = .175) in comparison with each respective control group. Furthermore, the median trabecular separation was significantly higher in the patients with Crohn's disease (7.4% higher, P = .048) but not in the patients with ulcerative colitis (6.3% higher, P = .217) compared with the respective controls.

Predictors of Bone Microstructure Deficiencies in Young Adult Men With Childhood-onset IBD

To determine whether having childhood-onset IBD per se was independently associated with the observed deficits in bone geometry and microstructure or dependent on differences in skeletal muscle index or physical exercise amount, we performed a multiple linear regression analysis for each HR-pQCT measurement variable (Table 4). Each analysis was adjusted for the SMI and the amount of physical exercise. A diagnosis of IBD was independently associated with a smaller cortical area but not with the total bone area or trabecular area (Table 4). Furthermore, IBD was independently associated with lower total vBMD, cortical vBMD, trabecular volume fraction and trabecular thickness, and thinner cortices (Table 4). However, we found no clear association between IBD diagnosis and the extent of trabecular separation, trabecular number, or periosteal circumference (Table 4).

Further, in these linear regression analyses, we found in both patients and controls that SMI was positively independently associated with the total, cortical, and trabecular area, as well as with total vBMD, but not with the cortical vBMD. Skeletal muscle index was also positively independently associated with all the measured cortical and trabecular bone microstructure parameters (Table 4). Furthermore, we found that physical exercise in hours/week in both diagnostic groups was not independently associated with bone geometric, vBMD, or microstructural measurements (Table 4). However, when performing the same analyses without adjusting for SMI, physical exercise was independently associated with higher cortical thickness but not with other microstructural measurements.

Discussion

In this study, young adult male patients with childhood-onset IBD were examined with HR-pQCT to evaluate their bone geometry and bone microstructure. For comparison, we used age- and height-matched young adult male controls from the same region. Our results reveal that young adult male patients with childhood-onset IBD display deficits in both the cortical and trabecular microstructures of the tibia.

Young men with childhood-onset IBD had a 12% lower 3D median vBMD than controls. In accordance, 2 recently published studies including both male and female patients, one with middle-aged participants and the other with young adults, showed that vBMD was lower in adults with IBD than in controls.^{18,19} The most striking deficit in our patients was the combination of a median of 17% thinner cortical layer and 6% thinner trabeculae compared with the controls. In contrast, previous studies using HR-pQCT for patients with



Table 2.	Bone geometry	and microstructure in young	adult males with childhood	d-onset inflammatory	bowel disease compared	with controls.
----------	---------------	-----------------------------	----------------------------	----------------------	------------------------	----------------

	Patients With IBD (n = 49)	Age- and Height-Matched Controls (n = 245)	Difference (%)	Р
Bone Geometry				
HR-pQCT Total area (mm ²)	844.8 (761.7-957.6)	849.7 (741.8-956.8)	-0.6	.697
HR-pQCT Cortical area (mm ²)	126.1 (110.9-147.1)	151.1 (131.9-171.9)	-16.5	<.001
HR-pQCT Trabecular area (mm ²)	706.9 (632.4-848)	692.3 (591.9-813.4)	2.1	.179
Volumetric Bone Mineral Density				
HR-pQCT vBMD (mg/cm ³)	296.7 (282.5-322.9)	336.7 (309.7-368.2)	-11.9	<.001
HR-pQCT Cortical BMD (mg/cm3)	854.4 (821-878.5)	878.5 (858.2-901.2)	-2.7	<.001
Bone Microstructure				
HR-pQCT Cortical thickness (mm)	1.110 (0.975-1.275)	1.330 (1.155-1.515)	-16.5	<.001
HR-pQCT Periosteal circumference (mm)	113.4 (107.9-121.0)	114.0 (107.0-121.5)	-0.5	.914
HR-pQCT Trabecular volumefraction (%)	16.8 (15.4-18.1)	18.2 (16.5-19.8)	-7.7	<.001
HR-pQCT Trabecular number (mm ⁻¹)	1.96 (1.75-2.27)	2.03 (1.83-2.255)	-3.4	.156
HR-pQCT Trabecular thickness (mm)	0.084 (0.076-0.089)	0.089 (0.083-0.096)	-5.6	<.001
HR-pQCT Trabecular separation (mm)	0.429 (0.364-0.488)	0.401 (0.357-0.451)	7.0	.036

IBD, Inflammatory bowel disease; XCT, Xtreme CT (High-resolution peripheral quantitative computed tomography); vBMD, volumetric bone mineral density. All values are presented as median (IQR). Difference between groups were tested with Mann–Whitney *U* test.

Table 3. Bone characteristics analyzed both with DXA and HR-pQCT, in young adult males with ulcerative colitis or Crohn's disease, compared with ageand height-matched controls.

	Ulcerative Colitis $(n = 32)$	Age-Height Matched Controls (<i>n</i> = 160)	Р	Crohn's Disease $(n = 17)$	Age-Height Matched Controls (<i>n</i> = 85)	Р
Bone Geometry						
HR-pQCT Total area (mm ²)	883.2 (807.2-963.2)	861.0 (744.0-965.5)	0.365	813.4 (709.3-940.2)	836.4 (736.8-944.6)	.539
HR-pQCT Cortical area (mm ²)	125.2 (113.3-150.2)	151.5 (134.2-172.8)	< 0.001	129.2 (108.5-145.7)	150.0 (125.4-168.9)	.001
HR-pQCT Trabecular area (mm ²)	732.8 (680.2-863.3)	701.8 (590.9-817.2)	0.097	666.0 (600.9-810.6)	686.6 (592.0-803.7)	.971
Volumetric Bone Mineral	Density					
HR-pQCT vBMD (mg/cm ³)	297.8 (278.2-325.2)	335.9 (307.4-369.9)	< 0.001	296.7 (284.9-317.8)	343.5 (310.6-363.5)	<.001
HR-pQCT Cortical BMD (mg/cm ³)	850.3 (824.8-865.2)	879.7 (862.2-902.8)	< 0.001	860.4 (810.9-896.9)	875.0 (851.3-897.3)	.175
Bone Microstructure						
HR-pQCT Cortical thickness (mm)	1.120 (0.945-1.263)	1.335 (1.170-1.520)	< 0.001	1.080 (0.975-1.305)	1.290 (1.140-1.490)	.007
HR-pQCT Periosteal circumference (mm)	116.2 (110.1-121.1)	114.8 (107.5-121.6)	0.515	111.9 (103.2-119.0)	113.7 (106.5-121.5)	.478
HR-pQCT Trabecular volumefraction (%)	17.4 (15.4-18.3)	18.1 (16.3-19.5)	0.008	16.4 (15.1-17.3)	18.6 (16.6-20.2)	<.001
HR-pQCT Trabecular number (mm ⁻¹)	2.015 (1.813-2.263)	2.050 (1.833-2.250)	0.435	1.930 (1.640-2.275)	2.010 (1.830-2.265)	.167
HR-pQCT Trabecular thickness (mm)	0.084 (0.076-0.089)	0.088 (0.081-0.095)	0.011	0.083 (0.075-0.089)	0.091 (0.084-0.099)	.002
HR-pQCT Trabecular separation (mm)	0.424 (0.363-0.466)	0.399 (0.356-0.451)	0.217	0.435 (0.367-0.519)	0.405 (0.358-0.451)	.048

Difference between groups were tested with Mann-Whitney U test. All values are presented as median (IQR).

Abbreviations: BMČ, bone mineral content; BMD, bone mineral density; BMI, body mass index; IBD, inflammatory bowel disease; SMI, skeletal muscle index; XCT, XtremeCT (High-resolution peripheral quantitative computed tomography).

IBD have reported deficits in either the cortical¹⁹ or the trabecular¹⁸ structures, but not in both. Deficiencies in either of these microstructures are reported to affect the bone strength.³⁰ We show that patients with ulcerative colitis and those with Crohn's disease have deficiencies in both their cortical and trabecular structures compared with controls. The deficiencies were most prominent in cortical area and vBMD Table 4. Multiple linear regression analyses showing the associations of having an IBD diagnosis and bone microstructure measurements, with adjustments for skeletal muscle index and physical exercise.

	Standardized Coefficients	Р	Unstandardized Coefficients	Standard Error	95.0% Confidence Interval for B	
	Beta		В		Lower Bound	Upper Bound
Total area (mm ²), adj. R ² (0.083)						
IBD vs Controls	0.06	.33	22.78	23.31	-23.10	68.65
Physical exercise (h/w)	-0.03	.57	-1.31	2.32	-5.87	3.25
SMI (kg/m^2)	0.31	<.001	50.96	9.54	32.18	69.74
Cortical area (mm ²), adj. R ² (0.24)	7)					
IBD vs Controls	-0.26	<.001	-20.41	3.98	-28.24	-12.59
Physical exercise (h/w)	0.05	.32	0.40	0.40	-0.38	1.17
SMI (kg/m ²)	0.39	<.001	11.85	1.63	8.65	15.05
Trabecular area (mm ²), adj. R ² (0.	049)					
IBD vs Controls	0.10	.08	42.68	24.08	-4.71	90.07
Physical exercise (h/w)	-0.04	.46	-1.77	2.39	-6.48	2.94
SMI (kg/m^2)	0.24	<.001	39.82	9.86	20.42	59.21
Volumetric BMD (mg/cm ³), adi. R	$R^{2}(0.137)$					
IBD vs Controls	-0.31	<.001	-39.99	6.98	-53.71	-26.26
Physical exercise (h/w)	0.05	.37	0.62	0.69	-0.75	1.98
SMI (kg/m ²)	0.17	<.001	8.46	2.86	2.84	14.08
Cortical BMD (mg/cm ³), adi, R ² (0.079)					
IBD vs Controls	-0.30	<.001	-27.69	5.23	-37.98	-17.40
Physical exercise (h/w)	0.02	.77	0.15	0.52	-0.87	1.17
SMI (kg/m ²)	-0.04	.51	-1.40	2.14	-5.61	2.82
Cortical thickness (mm), adi, R ² (().133)					
IBD vs Controls	-0.26	<.001	-0.20	0.04	-0.28	-0.12
Physical exercise (h/w)	0.07	.2.5	0.01	0.00	0.00	0.01
SMI (kg/m ²)	0.22	<.001	0.06	0.02	0.03	0.10
Periosteal circumference (mm), ad	i. R ² (0.092)					
IBD vs Controls	0.04	.45	1.18	1.55	-1.86	4.22
Physical exercise (h/w)	-0.03	.56	-0.09	0.15	-0.39	0.21
SMI (kg/m ²)	0.33	<.001	3.57	0.63	2.33	4.82
Trabecular volume fraction (%), a	di. R^2 (0.121)		0.07	0.00	2.00	
IBD vs Controls	-0.24	<.001	-0.02	0.00	-0.02	-0.01
Physical exercise (h/w)	0.00	.98	0.00	0.00	0.00	0.00
SMI (kg/m ²)	0.2.5	<.001	0.01	0.00	0.00	0.01
Trabecular number (mm ⁻¹), adi, R	² (0.159)		0101	0.000	0.00	0101
IBD vs Controls	-0.03	.57	-0.02	0.04	-0.11	0.06
Physical exercise (h/w)	-0.03	.60	0.00	0.00	-0.01	0.01
SMI (kg/m ²)	0.41	<.001	0.13	0.02	0.09	0.16
Trabecular thickness (mm), adi, R	² (0.070)				,	
IBD vs Controls	-0.25	<.001	-0.01	0.00	-0.01	0.00
Physical exercise (h/w)	0.04	.46	0.00	0.00	0.00	0.00
SMI (kg/m^2)	-0.17	<.001	0.00	0.00	0.00	0.00
Trabecular separation (mm), adi. 1	R^2 (0.158)		~~~ ~ ~		~~~ ~	~ ~ ~ ~
IBD vs Controls	0.08	.14	0.02	0.01	-0.01	0.04
Physical exercise (h/w)	0.01	.81	0.00	0.00	0.00	0.00
SMI (kg/m ²)	-0.40	<.001	-0.03	0.00	-0.04	-0.02

Abbreviations: CI, confidence interval; h/w, hours/week; IBD, inflammatory bowel disease; SMI, skeletal muscle index

in patients with IBD; this should not be due to bone size differences, as controls were matched for height and age. In contrast, Haschka et al¹⁹ reported such extensive deficits in patients with Crohn's disease only.

The widespread cortical and trabecular deficits seen in young adult men with childhood-onset IBD may be attributable to the fact that disease-onset occurred during childhood in our patients, thereby affecting a crucial phase of bone development. Other possible explanations for the discrepancies are our larger control cohort than previous studies and that we studied male patients in early adult-hood only. It is known that young adult males have thicker cortical³¹ and trabecular³² microstructures compared with young adult females, resulting in increased variation of bone parameters when analyzing both genders together. Moreover, recent studies using DXA have indicated that men with IBD may have more extensive BMD deficits than women with IBD, both early⁶ and later¹¹ in adulthood.

To determine if the changes in bone geometry and microstructure are independently associated with a diagnosis of IBD, we performed a series of multiple regression analyses. The results of these regression analyses confirmed that IBD is independently associated with deficits in both cortical and trabecular structures when adjustments are made for SMI and physical exercise. The IBD diagnosis per se entails several disease-specific risk factors for BMD deficits, such as chronic inflammation,³³ nutritional deficiency,³⁴ and prednisolone treatment.³⁵ These factors are also likely to affect the bone microstructure, especially the trabecular bone structures, which represent the more metabolically active part of the bone.³⁶

Our present findings indicate that young adult men with childhood-onset IBD have disturbances in bone microstructure leading to weaker bones, which may increase the risk of fracture.^{16,37} Interestingly, SMI was independently associated with thicker and denser cortical bone and a more robust trabecular structure, indicating the importance of skeletal muscle for structural bone development in patients with IBD. This supports the previous findings of Werkstetter and colleagues,³⁸ who measured the muscle cross-sectional area and bone geometry with pQCT in children with IBD. They reported positive correlations between gains in the muscle cross-sectional area, in both cortical and trabecular vBMD, and in cortical thickness.

The positive effects of physical exercise to BMD and bone microstructures are well known.³⁹ However in the current study, physical exercise was not independently associated with bone geometry or microstructure changes. Our research group has previously found a strong association between higher levels of physical exercise and increased SMI.²² This indicates that the association of physical exercise and bone microstructure is probably mediated through SMI, indicating the importance of building skeletal muscle mass in patients with IBD, so as to counteract the detrimental effects on the bone linked to the chronic inflammatory state.

The strengths of the present study are a relatively large cohort of young adult male patients with childhood-onset IBD in early adulthood and a robust control population that is matched for gender, age, and height. Furthermore, our cohort includes patients representing the whole clinical spectrum of childhood-onset IBD. A limitation of the study is that approximately one-third of the patients in our cohort were lost to follow-up. However, these patients did not clearly differ from the participants regarding diagnosis, age at diagnosis, medical regime, or gender; thus in our opinion, the present data remain representative for the cohort. Furthermore, in our multivariable regression models, all significant factors had a P value < .001. Therefore, we did not apply corrections for the multiple regression analysis, as this would not change the fact that our findings are statistically significant (P < .05).

To summarize, our research group has previously shown that our group of male patients in early adulthood with childhood-onset IBD have lower aBMD than normative matched controls.⁶ In the current study, we show that these young adult male patients with IBD are at increased risk of having compromised bone microstructure, with deficits of primarily in the cortical envelope but also in the trabecular bone compartment.

Acknowledgments

Authors thank all of their patients for participation in this study, Britt-Marie Käck (RN), and the Geriatric Medicine Clinic staff in Gothenburg, especially Senada Catic, Ulrika Hjertonsson, and Vera Gligoric, for the measurements of our patients.

Author Contributions

G.V.S, R.S., and M.L. contributed to the study design, data collection, data analysis, and manuscript writing.

S.S., D.M., and C.O. contributed to the study design, data collection, and manuscript review.

All the authors approved the final draft of the manuscript.

Funding

ALF research funding; The Sahlgrenska Academy at the University of Gothenburg and Region Västra Götaland, Sweden; Queen Silvia Children's Hospital's Research Fund, Gothenburg, Sweden; The Inga-Britt and Arne Lundberg Foundation.

Conflicts of Interest

The authors have no conflicts of interest to disclose.

References

- Schmidt S, Mellstrom D, Norjavaara E, Sundh V, Saalman R. Longitudinal assessment of bone mineral density in children and adolescents with inflammatory bowel disease. J Pediatr Gastroenterol Nutr. 2012;55(5):511-518.
- Schmidt S, Mellstrom D, Norjavaara E, Sundh SV, Saalman R. Low bone mineral density in children and adolescents with inflammatory bowel disease: a population-based study from Western Sweden. *Inflamm Bowel Dis.* 2009;15(12):1844-1850.
- Sylvester FA, Wyzga N, Hyams JS, et al. Natural history of bone metabolism and bone mineral density in children with inflammatory bowel disease. *Inflamm Bowel Dis.* 2007;13(1):42-50.
- Laakso S, Valta H, Verkasalo M, Toiviainen-Salo S, Viljakainen H, Makitie O. Impaired bone health in inflammatory bowel disease: a case-control study in 80 pediatric patients. *Calcif Tissue Int.* 2012;91(2):121-130.
- Gokhale R, Favus MJ, Karrison T, Sutton MM, Rich B, Kirschner BS. Bone mineral density assessment in children with inflammatory bowel disease. *Gastroenterology*. 1998;114(5):902-911.
- Sigurdsson GV, Schmidt S, Mellstrom D, et al. Bone mass development from childhood into young adulthood in patients with childhood-onset inflammatory bowel disease. *Inflamm Bowel Dis.* 2017;23(12):2215-2226.
- Laakso S, Valta H, Verkasalo M, Toiviainen-Salo S, Makitie O. Compromised peak bone mass in patients with inflammatory bowel disease—a prospective study. *J Pediatr.* 2014;164(6):1436-43.e1.

- Guz-Mark A, Rinawi F, Egotubov O, Shimon I, Shamir R, Assa A. Pediatric-onset inflammatory bowel disease poses risk for low bone mineral density at early adulthood. *Dig Liver Dis*. 2017:49(6):639-642.
- Schule S, Rossel JB, Frey D, et al. Prediction of low bone mineral density in patients with inflammatory bowel diseases. United European Gastroenterol J. 2016;4(5):669-676.
- Schulte C, Dignass AU, Mann K, Goebell H. Reduced bone mineral density and unbalanced bone metabolism in patients with inflammatory bowel disease. *Inflamm Bowel Dis.* 1998;4(4):268-275.
- 11. Walldorf J, Krummenerl A, Engler K, et al. Health care for osteoporosis in inflammatory bowel disease: unmet needs in care of male patients? *Journal of Crohn's & colitis*. 2013;7(11):901-907.
- Azzopardi N, Ellul P. Risk factors for osteoporosis in Crohn's disease: infliximab, corticosteroids, body mass index, and age of onset. *Inflamm Bowel Dis.* 2013;19(6):1173-1178.
- 13. Targownik LE, Bernstein CN, Nugent Z, Leslie WD. Inflammatory bowel disease has a small effect on bone mineral density and risk for osteoporosis. *Clin Gastroenterol Hepatol*. 2013;11(3):278-285.
- 14. Johnell O, Kanis JA, Oden A, et al. Predictive value of BMD for hip and other fractures. *J Bone Miner Res.* 2005;20(7):1185-1194.
- Targownik LE, Bernstein CN, Leslie WD. Inflammatory bowel disease and the risk of osteoporosis and fracture. *Maturitas*. 2013;76(4):315-319.
- Ohlsson C, Sundh D, Wallerek A, et al. Cortical bone area predicts incident fractures independently of areal bone mineral density in older men. J Clin Endocrinol Metab. 2017;102(2):516-524.
- 17. Chapurlat R, Bui M, Sornay-Rendu E, et al. Deterioration of cortical and trabecular microstructure identifies women with osteopenia or normal bone mineral density at imminent and long-term risk for fragility fracture: a prospective study. J Bone Miner Res. 2020;35(5):833-844.
- Pepe J, Zawadynski S, Herrmann FR, et al. Structural basis of bone fragility in young subjects with inflammatory Bowel Disease: a high-resolution pQCT study of the SWISS IBD Cohort (SIBDC). *Inflamm Bowel Dis.* 2017;23(8):1410-1417.
- Haschka J, Hirschmann S, Kleyer A, et al. High-resolution quantitative computed tomography demonstrates structural defects in cortical and trabecular bone in IBD patients. *Journal of Crohn's & colitis.* 2016;10(5):532-540.
- Sigurdsson GV, Schmidt S, Mellstrom D, et al. Altered body composition profiles in young adults with childhood-onset inflammatory bowel disease. *Scand J Gastroenterol*. 2020;55(2):169-177.
- Nobile S, Grand RJ, Pappa HM. Risk factors for low bone mineral density in pediatric inflammatory bowel disease: the positive role of physical activity. *Eur. J. Gastroenterol. Hepatol.* 2018;30(4): 471-476.
- 22. Sigurdsson GV, Schmidt S, Mellstrom D, et al. Physical exercise is associated with beneficial bone mineral density and body composition in young adults with childhood-onset inflammatory bowel disease. *Scand J Gastroenterol.* 2021;56(6):699-707.
- Darelid A, Nilsson M, Kindblom JM, Mellström D, Ohlsson C, Lorentzon M. Bone turnover markers predict bone mass development in young adult men: A five-year longitudinal study. J Clin Endocrinol Metab. Article. 2015;100(4):1460-1468.

- 24. Satsangi J, Silverberg MS, Vermeire S, Colombel JF. The Montreal classification of inflammatory bowel disease: controversies, consensus, and implications. *Gut.* 2006;55(6):749-753.
- Ruemmele FM, Veres G, Kolho KL, et al. Consensus guidelines of ECCO/ESPGHAN on the medical management of pediatric Crohn's disease. J Crohn's colitis. 2014;8(10):1179-1207.
- 26. Turner D, Levine A, Escher JC, et al. Management of pediatric ulcerative colitis: joint ECCO and ESPGHAN evidence-based consensus guidelines. *J Pediatr Gastroenterol Nutr.* 2012;55(3): 340-361.
- 27. Sundh D, Mellstrom D, Ljunggren O, et al. Low serum vitamin D is associated with higher cortical porosity in elderly men. *J Intern Med.* 2016;280(5):496-508.
- Laib A, Hauselmann HJ, Ruegsegger P. In vivo high resolution 3D-QCT of the human forearm. *Technol Health Care*. 1998;6(5-6):329-337.
- 29. Rudang R, Darelid A, Nilsson M, et al. Smoking is associated with impaired bone mass development in young adult men: a 5-year longitudinal study. J Bone Miner Res. 2012;27(10):2189-2197.
- 30. Pistoia W, van Rietbergen B, Ruegsegger P. Mechanical consequences of different scenarios for simulated bone atrophy and recovery in the distal radius. *Bone*. 2003;33(6):937-945.
- 31. Kazakia GJ, Nirody JA, Bernstein G, Sode M, Burghardt AJ, Majumdar S. Age- and gender-related differences in cortical geometry and microstructure: Improved sensitivity by regional analysis. *Bone.* 2013;52(2):623-631.
- 32. Sode M, Burghardt AJ, Kazakia GJ, Link TM, Majumdar S. Regional variations of gender-specific and age-related differences in trabecular bone structure of the distal radius and tibia. *Bone*. 2010;46(6):1652-1660.
- Lopes LH, Sdepanian VL, Szejnfeld VL, de Morais MB, Fagundes-Neto U. Risk factors for low bone mineral density in children and adolescents with inflammatory bowel disease. *Dig Dis Sci.* 2008;53(10):2746-2753.
- Hartman C, Eliakim R, Shamir R. Nutritional status and nutritional therapy in inflammatory bowel diseases. World J Gastroenterol. 2009;15(21):2570-2578.
- 35. LeBlanc CM, Ma J, Taljaard M, et al. Incident vertebral fractures and risk factors in the first three years following glucocorticoid initiation among pediatric patients with rheumatic disorders. J Bone Miner Res. 2015;30(9):1667-1675.
- Datta HK, Ng WF, Walker JA, Tuck SP, Varanasi SS. The cell biology of bone metabolism. J Clin Pathol. 2008;61(5):577-587.
- 37. Samelson EJ, Broe KE, Xu H, et al. Cortical and trabecular bone microarchitecture as an independent predictor of incident fracture risk in older women and men in the Bone Microarchitecture International Consortium (BoMIC): a prospective study. *Lancet Diabetes Endocrinol.* 2019;7(1):34-43.
- Werkstetter KJ, Pozza SB, Filipiak-Pittroff B, et al. Long-term development of bone geometry and muscle in pediatric inflammatory bowel disease. *Am J Gastroenterol.* 2011;106(5):988-998.
- 39. Nilsson M, Ohlsson C, Oden A, Mellstrom D, Lorentzon M. Increased physical activity is associated with enhanced development of peak bone mass in men: a five-year longitudinal study. J Bone Miner Res. 2012;27(5):1206-1214.